



B The Brookings Institution

The Hamilton Project seeks to advance America's promise of opportunity, prosperity, and growth. The Project's economic strategy reflects a judgment that long-term prosperity is best achieved by making economic growth broad-based, by enhancing individual economic security, and by embracing a role for effective government in making needed public investments. Our strategy—strikingly different from the theories driving economic policy in recent years—calls for fiscal discipline and for increased public investment in key growth-enhancing areas. The Project will put forward innovative policy ideas from leading economic thinkers throughout the United States—ideas based on experience and evidence, not ideology and doctrine—to introduce new, sometimes controversial, policy options into the national debate with the goal of improving our country's economic policy.

The Project is named after Alexander Hamilton, the nation's first treasury secretary, who laid the foundation for the modern American economy. Consistent with the guiding principles of the Project, Hamilton stood for sound fiscal policy, believed that broad-based opportunity for advancement would drive American economic growth, and recognized that "prudent aids and encouragements on the part of government" are necessary to enhance and guide market forces.





Getting More from Low-Income Housing Assistance

Edgar O. Olsen University of Virginia

NOTE: This discussion paper is a proposal from the author. As emphasized in The Hamilton Project's original strategy paper, the Project was designed in part to provide a forum for leading thinkers across the nation to put forward innovative and potentially important economic policy ideas that share the Project's broad goals of promoting economic growth, broad-based participation in growth, and economic security. The authors are invited to express their own ideas in discussion papers, whether or not the Project's staff or advisory council agrees with the specific proposals. This discussion paper is offered in that spirit.

The Brookings Institution

SEPTEMBER 2008

Abstract

This paper argues that the two most serious structural shortcomings of the current system of low-income housing assistance are (1) its excessive reliance on unit-based assistance and (2) its failure to provide housing assistance to all of the poorest eligible families who ask for help. Evidence on the performance of housing programs indicates that unit-based assistance has a much greater cost than tenant-based assistance for providing equally good housing, and it needlessly restricts recipient choice. Unit-based assistance has no advantage over tenant-based assistance to offset these disadvantages. The nonentitlement nature of the current system is inconsistent with plausible assumptions about taxpayer preferences. The paper argues for a transition to an entitlement housing assistance program that relies exclusively on tenant-based assistance. It describes concrete actions that would achieve this result without spending additional money, and it shows that the major objections to these proposals are inconsistent with the evidence on program performance. The proposed transition would benefit most current recipients of housing assistance, and the reforms would give those taxpayers who want to help low-income families with their housing more for their money. After the transition is complete, millions of additional families would receive housing assistance that enables them to occupy better housing in nicer neighborhoods, and to consume more of other goods. Millions of other families that would have received unit-based assistance with the continuation of the current system would live in housing, neighborhoods, and locations that they prefer to their units in subsidized projects.

Copyright © 2008 The Brookings Institution

Contents

1. Introduction	5
2. Overview of Current System of Low-Income Housing Assistance	7
3. Comparing Unit-Based and Tenant-Based Housing Assistance	9
4. Proposals to Shift Budget from Unit-Based to Tenant-Based Assistance	17
5. Justification and Design of an Entitlement Housing Voucher Program	24
6. Major Effects of Proposed Reforms	30
7. Objections to Exclusive Reliance on Tenant-Based Assistance	31
8. Conclusion	36
References	37

1. Introduction

ow-income housing programs are an important part of the U.S. welfare system. The most widely cited figure for government expenditure on these programs, about \$30 billion a year, refers to the U.S. Department of Housing and Urban Development's (HUD's) direct expenditure. This figure ignores the large and rapidly growing Low-Income Housing Tax Credit (LIHTC), major U.S. Department of Agriculture (USDA) programs, expenditures of state and local governments, and the many indirect subsidies that account for a large part of the cost of the system. In fact, governments in the United States directly or indirectly spend roughly \$50 billion a year on low-income housing programs. This means that they spend substantially more on housing subsidies to the poor than on other better-known parts of the welfare system such as Temporary Assistance for Needy Families (TANF) and Food Stamps.

The purpose of this paper is to describe the major structural shortcomings of the current system of low-income rental housing assistance and to propose how these shortcomings can be remedied without spending more money. The most serious structural shortcomings of the current system are its excessive reliance on unit-based programs that serve about two-thirds of assisted households, and its failure to provide housing assistance to all of the poorest eligible families who ask for help. Evidence indicates that tenant-based housing vouchers have a much lower total cost than any program of unitbased assistance for providing equally good housing. Therefore, by shifting resources from unit-based to tenant-based assistance it would be possible to serve current recipients equally well (that is, provide them with equally good housing for the same rent) and serve many additional families without spending more money. This would involve terminating or phasing out current production programs, disengaging from unit-based assistance to existing apartments as soon as current contractual commitments permit, and avoiding new programs of unit-based assistance. The savings from these actions would make it possible to create an entitlement housing assistance program serving millions of additional households without spending more money, thereby avoiding the inequity of providing assistance to some households and denying it to other similar households.

This paper does not address the issue of how much to spend on low-income housing assistance. Instead, it deals with how to deliver any amount of assistance—whether more, less, or the same amount as the current system. The desirability of the proposed reforms does not depend on how much is spent on low-income housing assistance. If more money is spent on it, more families will be helped and the families assisted will receive larger benefits.

The paper also does not deal with the fundamental issue of whether housing assistance to low-income households is preferable to unrestricted cash grants for these households. Several coherent arguments for housing assistance have been offered (Olsen 2003, pp. 368-70). For example, many taxpayers want to help low-income households but think that some low-income households undervalue housing and therefore might not provide adequate housing for themselves or their children. The proposals in this paper are based on the assumption that it is desirable to induce at least some low-income families to occupy housing that is better in at least certain respects than the housing that they would occupy if they were given cash grants with no strings attached, even though the recipients themselves would prefer the unrestricted cash grants.

The paper is organized as follows. Section 2 provides an overview of the current system of low-income housing assistance. Section 3 summarizes the most important evidence on the performance of different rental housing programs, presents evidence

on the magnitude of the increase in the number of households that could be served with the current budget for low-income housing assistance by shifting from unit-based to tenant-based assistance, and discusses household choice, the other major advantage of tenant-based housing assistance. Section 4 describes concrete proposals for phasing out unit-based housing assistance. Section 5 argues that tax-payer preferences call for an entitlement housing assistance program for the poorest eligible families and shows how this can be achieved without spending more money. Section 6 summarizes the major effects of the proposed reforms. Section 7 addresses the main objections to exclusive reliance on tenant-based assistance. Section 8 concludes the paper.

2. Overview of Current System of Low-Income Housing Assistance

he U.S. government provides assistance to some households who live in rental and owner-occupied housing.1 The most important distinction between rental housing programs is whether the subsidy is attached to the dwelling unit or the assisted household. If the subsidy is attached to a rental dwelling unit, each family must accept the particular unit offered in order to receive assistance and loses the subsidy when it moves. Each family offered tenant-based rental assistance has a choice among many units that meet the program's standards, and the family can retain its subsidy when it moves. Homeownership programs fall into two analogous categories. Some programs authorize selected developers to build a limited number of houses to sell to eligible families of their choosing and require eligible families to buy from these builders in order to receive a subsidy. Other homeownership programs provide subsidies to eligible families that are free to buy from any seller whose unit meets the program's housing standards.

Unit-based rental assistance is the dominant form of direct federal housing assistance to low-income families. The overwhelming majority of housing assistance recipients receives rental assistance, and more than 70 percent of these renting families receive unit-based assistance. HUD provides unit-based rental assistance to about 2.7 million families, LIHTC projects house about 1.6 million families, and the USDA's Section 515/521 and HUD's HOME block grant program each serve almost a half million families in subsidized projects. HUD's Section 8 Voucher program, which accounts for almost all tenant-based rental housing assistance in the United States, serves about 2 million house-holds.

There are two broad types of unit-based rental as-

sistance: (1) public housing and (2) privately owned subsidized projects. Public housing projects are owned and operated by local public housing authorities established by local governments. The overwhelming majority of projects were newly built for the program. Until 1969, with minor exceptions, federal taxpayers paid the initial development cost of public housing while tenants and local taxpayers paid the operating cost. However, the federal government now provides local housing authorities with substantial operating and modernization subsidies. In the public housing program, local and federal government officials and employees make all of the decisions that are made by private owners in the unsubsidized housing market.

The federal government also has contracted with private parties to provide unit-based assistance in subsidized housing projects. The majority of these private parties have been for-profit firms, but nonprofit organizations also have had a significant presence. The largest programs of this type are the IRS's LIHTC, HUD's Section 8 New Construction and Substantial Rehabilitation programs and Section 236 Rental and Cooperative Housing for Lower-Income Families programs, and USDA's Section 515/521. Under most programs, these private parties agree to provide rental housing meeting certain standards at restricted rents to households with particular characteristics for a specified number of years. The overwhelming majority of the projects have been newly built under a subsidized construction program. Almost all of the rest were substantially rehabilitated as a condition for participation in the program.

It is important to realize that none of these programs provides subsidies to all suppliers who would like to participate. Since subsidies are provided to

^{1.} See Olsen (2003, pp. 370–94) for a more detailed description of the system of low-income rental housing programs, and Olsen (2007) for a more detailed account of homeownership programs that serve low-income households.

selected private suppliers, competition among these suppliers does not ensure that subsidies will be passed along to occupants of the subsidized units in full, or indeed at all. The only way to ensure that subsidies are passed down is through administrative mechanisms. Without these mechanisms, for-profit suppliers would charge rents and provide a level of services so that the marginal occupant of the project would receive no net benefit from the program and no additional eligible families would want to live in the project. Furthermore, these suppliers would receive a higher return on their investment of time and money than they could earn in the unsubsidized market.

The federal government has administered two large homeownership programs for low-income households: (1) USDA's Section 502 Single Family Direct Loan Housing program, and (2) HUD's Section 235 Homeownership program. Section 502 provides assistance directly to homeowners for houses of their own choice. Section 235 has not made new commitments for the past two decades, but continues to provide subsidies on behalf of a small number of families. Its larger component provided assistance tied to construction of particular dwelling units. The

smaller existing housing component used a procedure for allocating subsidies that had elements of unit-based and tenant-based assistance. Over their histories, these programs have subsidized housing for about 2.5 million families.²

The federal government also provides funds to state and local governments intended to subsidize the housing of low-income families. The HOME Investment Partnerships program is a housing block grant that has been used to provide rental and homeownership subsidies. Under the Mortgage Revenue Bond program, state housing agencies issue bonds whose interest is not subject to the federal individual income tax and use the proceeds to provide below-market rate loans to low-income, first-time homebuyers. The smaller Multifamily Housing Bond program provides below-market rate loans for the construction of rental housing projects for low-income households. Finally, about one-fourth of the funding of the Community Development Block Grant program is used to provide housing assistance to low- and moderate-income families. State and local governments operate programs of unit-based and tenant-based assistance with the funding from these programs.

According to Mikesell and colleagues (1999, p. 3), Section 502 had served about 1.9 million families through 1998. Carliner (1998, pp. 313–14) reports that about 525,000 households received Section 235 subsidies over its history.

3. Comparing Unit-Based and Tenant-Based Housing Assistance

he most important finding of the empirical literature on the performance of low-income housing programs from the viewpoint of housing policy is that tenant-based housing assistance has provided equally good housing at a much lower cost than any type of unit-based assistance.³ Given these cost savings, a nationwide tenant-based assistance program would serve at least 20 percent and more likely 70 percent more families than today's unit-based assistance program, without spending more money. In addition, a tenant-based housing program would afford eligible families more housing choice, allowing them to choose a home more suited to their specific tastes and circumstances.

Evidence on Cost-Effectiveness

The best cost-effectiveness analyses of housing programs involve a comparison of the total cost of providing the housing with its *market rent* (an index of the overall desirability of the dwelling). For tenant-based vouchers and certificates, the approach is straightforward. All costs associated with providing the housing during a period occur in that period, and all those costs are in the records of the administering agency. To estimate the market rents of the units occupied by subsidized households, analysts first estimate a statistical relationship between the rent and characteristics of unsubsidized apartments, and then substitute the characteristics of the subsidized units into this estimated equation.

Dealing with construction or rehabilitation programs is more difficult because the time path of cost

bears no particular relationship to the time path of the market rent of a unit and all of these programs involve indirect costs that are not in the records of the administering agency. The most widely accepted measure of cost-effectiveness for programs of this type is the ratio of A to B, where A is the present value of the rents paid by tenants and all direct and indirect costs incurred by federal, state, and local governments and B is the present value of the market rents of the units over the period that the units are used to house subsidized families. If a government owns the project at the time that it stops being used to house subsidized families, the present value of the project's market value at that time should be subtracted from the present value of the costs.

Four major studies have estimated both the cost per unit and the mean market rent of apartments provided by tenant-based housing certificates and vouchers and the largest older production programs, namely, Public Housing, Section 236, and the Section 8 New Construction program.⁵ These studies are based on data from a wide variety of housing markets and for projects built in many different years. Two are detailed studies conducted for HUD by a respected research firm during the Nixon, Ford, Carter, and Reagan administrations. The four studies are unanimous in finding that housing certificates and vouchers provide equally desirable housing at a much lower total cost than any of these production programs do, even though all of these studies are biased in favor of the production programs, to some extent, by the omission of certain indirect costs.

^{3.} Other aspects of program performance have been studied. In some cases, the differences between the estimated effects of different programs are small. In other cases, the differences could be eliminated without fundamental modifications of the programs involved. For example, if recipients of housing vouchers occupied better housing, on average, than public housing tenants, then this difference could be eliminated without spending more money by reducing the generosity of the voucher subsidy and devoting the savings to greater subsidies to housing authorities for the maintenance and renovation of their public housing projects. Olsen (2008) provides a brief summary of the evidence, and Olsen (2003, pp. 394–427) provides a more detailed account.

^{4.} Alternative indices are possible, but have not been used in cost-effectiveness analyses. For almost a decade, HUD's Real Estate Assessment Center has produced an overall index of the physical condition of units in each of HUD's subsidized projects. Unlike market rent, this index does not cover all aspects of the unit of value to occupants such as the quality of local public schools and the convenience of the location to jobs, shopping, and recreation. However, it does cover many important matters in great detail. This index has not been used to compare the cost-effectiveness of different methods of delivering housing assistance.

Table 1 summarizes the results of these studies. The studies with the most-detailed information about the characteristics of the housing provided by the programs found the largest excess costs for the production programs. Specifically, Mayo et al. (1980) estimate the excessive cost of public housing compared to housing vouchers for providing equally desirable housing to be 64 percent and 91 percent, respectively, in the two cities studied (Phoenix and Pittsburgh) and the excessive cost of Section 236 to be 35 percent and 75 percent, respectively, in those two cities. Another study with excellent data on housing characteristics estimates the excessive cost of the Section 8 New Construction program compared to the tenant-based Section 8 Certificates program to be between 44 percent and 78 percent (Wallace et al. 1981).6

Recent evidence supports these older findings. Several studies (Leger and Kennedy 1990; Mayo et al. 1980; Wallace et al. 1981; Weinberg 1982) have found that the total cost of various types of tenantbased housing assistance have exceeded the market rent of the units involved by only the cost of administering the program. In contrast, the total cost of the Section 8 New Construction and Substantial Rehabilitation programs, HUD's largest program that subsidized the construction of privately owned projects, has exceeded the market rents of the units by much more than the program's administrative cost. Based on a large random sample of HUD-insured Section 8 programs, Finkel and colleagues (1999, Exhibit 5–1) report that property owners received payments from tenants and the government in 1995 that exceeded market rent for 86 percent

TABLE 1
Excess Cost of Older Production Programs for Equally Good Housing

Program and study	Localities	Projects built (years)	Excess cost (percent)
Public Housing			
Olsen and Barton (1983)	New York City	1937–1965	14
Olsen and Barton (1983)	New York City	1937–1968	10
HUD (1974)	Baltimore, Boston, Los Angeles, St. Louis, San Francisco, Washington (DC)	1953–1970	17
Mayo et al. (1980)	Phoenix	1952–1974	64
Mayo et al. (1980)	Pittsburgh	1952–1974	91
Section 236			
Mayo et al. (1980)	Phœnix	1969–1975	35
Mayo et al. (1980)	Pittsburgh	1969–1975	75
Section 8 New Construction and Substantial Rehabilitation programs			
Wallace et al. (1981)	National	1979	44–78

Source: Author's calculations.

^{5.} The studies are HUD (1974), Mayo et al. (1980), Olsen and Barton (1983), and Wallace et al. (1981). Olsen (2000) provides a description and critical appraisal of the data and methods used in these studies as well as a summary of their results.

^{6.} This study made predictions of the market rents of subsidized units based on two different data sets including information on the rent and characteristics of unsubsidized units. The study did not collect information on the indirect costs of the Section 8 New Construction Program. These indirect subsidies include Government National Mortgage Association (GNMA, or Ginnie Mae) Tandem Plan interest subsidies for Federal Housing Administration—insured (FHA-insured) projects and the forgone tax revenue due to the tax-exempt status of interest on the bonds used to finance state housing finance agency projects. Based on previous studies, the authors argue that these indirect costs would add 20 to 30 percent to the total cost of the Section 8 New Construction Program. The range of estimates reported in the text is based on the four combinations of the two predictions of market rent, and on the lower and upper limits on the indirect costs.

of their units. Two-thirds of the units received payments that exceeded market rent by more than 20 percent. But even these calculations understate the excess cost for two reasons: (1) The predicted market rents are based on the assumption that all property systems are restored to their original working condition and hence overstate the market rents of the units in their current condition. (2) They ignore other subsidies received by these projects, such as the Ginnie Mae Tandem Plan interest subsidies received by all projects and the local property tax exemptions received by many projects. Shroder and Reiger (2000) produce similar results in a direct comparison of the total payments to property owners under the Section 8 Certificate program, and the Section 8 New Construction and Substantial Rehabilitation programs, finding that Section 8 projects are about 35 percent more expensive than certificates.

Recent U.S. General Accounting Office (GAO) studies have produced similar results for the major active construction programs-LIHTC, HOPE VI, Section 202, Section 515, and Section 811. Table 2 reports results based on the conceptually preferable life cycle approach.7 The excess total cost estimates range from 12 percent for Section 811 to 27 percent for HOPE VI (GAO 2001, p. 3). These estimates are lower bounds on the excessive cost because some costs of the production programs were omitted. Most notably, the opportunity cost of the land and cost of preparing the site were omitted from the cost of HOPE VI projects. These are real costs to society of HOPE VI redevelopment. More generally, some costs of each production program were omitted. For example, some projects under each program receive local property tax abatements. The preceding results ignore this cost to local taxpayers. Other projects are built on land sold to the developer by a government at a below-market price.

TABLE 2
Excess Cost of Active Production Programs

Program	Excess cost (percent)
LIHTC	16
Hope VI	27
Section 202	19
Section 811	12
Section 515	25

Source: GAO 2001.

It is often argued that production programs work better than tenant-based vouchers in the tightest housing markets. The GAO study includes evidence concerning whether production programs are more cost-effective than tenant-based vouchers in housing markets with low vacancy rates. In 1999, the rental vacancy rates in the seven metropolitan areas studied ranged from 3.1 percent in Boston to 7.2 percent in Baltimore and Dallas, meaning that all of the specific markets studied were tighter than the U.S. average of 7.8 percent. In each market, tenant-based vouchers were more cost-effective than each production program studied. Table 3 reports the results for Tax Credit program. The results for Section 202 and 811 are similar (GAO 2002, pp. 19-20).

^{7.} The GAO study also reports first-year excess costs of the production programs. The first-year cost of a production program is the sum of the annualized development subsidies and the tenant rent and other government subsidies during the first year of operation. The GAO estimates of excessive cost of production programs based on this method are much higher than estimates based on the life-cycle approach. Olsen (2000, pp. 18–21) explains the shortcomings of first-year-cost methodology and how this approach can bias the results in either direction.

TABLE 3
Excess Cost of Tax Credits in Markets with Different Vacancy Rates

Metropolitan area	Vacancy rate (all units) (percent)	One bedroom (percent)	Two bedroom (percent)
Baltimore	7.2	24	24
Boston	3.1	6	19
Chicago	6.5	34	25
Dallas/Fort Worth	7.2	21	21
Denver	5.6	40	21
Los Angeles	5.1	11	21
New York	4.7	21	17
All metro areas	7.8	19	14

Source: GAO 2002.

Unlike most of the earlier cost-effectiveness studies, the GAO study does not compare the total cost of dwellings under the different programs that were the same with respect to many characteristics. Instead, it simply compares the average cost of dwellings that have the same number of bedrooms in the same metropolitan area or the same type of location (metropolitan or nonmetropolitan). This has led to the criticism that the results overstate the excessive costs of the production programs for providing equally desirable housing because these programs provide better housing than the units occupied by voucher recipients. However, evidence from earlier construction programs casts doubt on this view. This evidence does not indicate a significant difference in overall desirability between units in recently completed projects under construction programs and units occupied by households with certificates and vouchers. Moreover, the relevant issue is not the quality of the housing under a construction program when it is new but rather the average quality of housing provided over the time that the project serves assisted households. This quality has typically declined over time, at least until additional subsidies are provided for major renovation.

In fact, the existing evidence suggests that, well before the units in subsidized projects reach the midpoint of their service to assisted households, they provide housing worse than the housing occupied by recipients of tenant-based vouchers and certificates. Mayo et al. (1980) estimate the market rents of units under several housing programs in Pittsburgh and Phoenix in 1975 based on data on the market rent and numerous characteristics of unsubsidized units and their neighborhoods. At the time of the study, none of these units had reached the midpoint of their useful lives. The results in Table 4 indicate that these public housing units were no better than the units occupied by recipients of housing allowances. In addition, the Section 236 units, which were only a few years old at the time, were not much better than the units occupied by recipients of housing allowances. In a study of sixteen randomly selected metropolitan areas in 1979, Wallace et al. (1981) find a difference of less than 10 percent in the estimated market rents of Section 8 existing units and Section 8 New Construction program units.

A similar picture emerges from David Vandenbroucke's unpublished tabulations based on the 1991 American Housing Survey Metropolitan Sample. (Vandenbroucke is with the Office of Policy Development and Research, HUD.) He estimates separate statistical relationships between market rent and numerous characteristics of unsubsidized units and their neighborhoods in a number of metropolitan areas and then uses these relationships to predict the market rents of public housing units, units in privately owned HUD-subsidized projects, and units occupied by certificate and voucher hold-

ers. Table 5 reports his results. In eight of eleven metropolitan areas the median market rent of the units occupied by recipients of certificates and vouchers was greater than the median market rent of units in public and privately owned HUD-subsidized projects. Based on the median age of public housing units at the time, it is plausible to believe the majority of public housing units in his sample had not reached the midpoint of their service to assisted households and that the majority of privately owned projects were much younger than public housing projects.

TABLE 4

Market Rents of Units under Production Programs in Their Early Years Compared with Voucher Units

City		Program	
	Section 236	Public housing	Housing allowance
Pittsburgh	\$1,826	\$1,748	\$1,626
Phoenix	\$2,417	\$1,918	\$2,084

Source: Mayo et al. (1980).

TABLE 5
Median Monthly Market Rents of Subsidized Units, 1991

		Program	
City	Voucher and certificate	Privately owned projects	Public housing
Atlanta	\$505	\$400	\$328
Baltimore	\$460	\$458	\$373
Chicago	\$475	\$550	\$440
Columbus	\$375	\$395	\$340
Hartford	\$593	\$570	\$543
Houston	\$365	\$325	N/A
New York	\$605	\$578	\$520
Newark	\$568	\$570	\$500
San Diego	\$480	\$410	N/A
Seattle	\$475	\$455	\$445
St. Louis	\$403	\$378	\$380

Source: David Vandenbroucke's (Office of Policy Development and Research, HUD) unpublished tabulations. N/A = not applicable.

The GAO study will not be the last word on the cost-effectiveness of the programs studied. Data on the condition, amenities, and neighborhoods of subsidized units over time are essential for accurately assessing the cost-effectiveness of alternative methods of delivering housing assistance. Other improvements in the GAO's implementation of the life cycle methodology are possible and desirable. However, the GAO study provides the only independent cost-effectiveness analysis of active subsidized construction programs.

The preceding evidence on the cost-effectiveness of project-based assistance applies to units built or substantially rehabilitated under a subsidized construction program and still under their initial use agreement. In contrast, the Mark-to-Market program provides evidence on the cost-effectiveness of renewing use agreements for privately owned subsidized projects, and the Experimental Housing Allowance program (EHAP) provides evidence on the cost-effectiveness of project-based assistance to existing, previously unsubsidized housing.

Congress created the Mark-to-Market program in the 1997 Multifamily Assisted Housing Reform and Affordability Act to reduce the excess subsidies under the project-based Section 8 programs. Early evidence on the experience with this initiative indicates that it failed to eliminate the excessive cost, however. Only one-third of the projects that had been restructured by July 31, 2003 received the market rent with no additional subsidies (Hilton et al. 2004, p. xiii). HUD repaid all or a part of the outstanding balance on the mortgages of the remaining projects. For more than one-fourth of these projects, HUD also agreed to monthly payments in excess of market levels. For 35 percent of the projects with above-market rents, the initial monthly payments exceeded market rents by more than 20 percent (Hilton et al., p. 40). Furthermore, we can expect the excess of the cost relative to market rent to increase over time because the projects receive automatic rent increases that are independent of the condition of their units.

EHAP provides additional evidence on the cost-effectiveness of project-based assistance. One type of housing allowance tested in EHAP was essentially identical to the housing voucher program that operated between 1983 and 1998: it offered each eligible family a subsidy that depended on the family's characteristics, on the condition that the family occupies a unit meeting minimum housing standards. At the time of EHAP, HUD operated the national Section 23 Existing Housing program, the first program of tenant-based rental housing assistance in the United States. Under one variant of this program, housing authorities rented existing apartments and sublet them to eligible families. This is analogous to the project-based component of the Section 8 Voucher program.

The results for one of the metropolitan areas studied provide clear evidence on the cost-effectiveness of tenant-based versus project-based assistance for existing housing (Mayo et al. 1980, pp. 134–39). All Section 23 units in Pittsburgh were leased by the housing authority and sublet to tenants. The ratio of total cost to market rent for these units was 1.67. For example, it cost \$835 to rent a unit with a market rent of \$500. The ratio for the tenant-based housing allowance program was 1.15.8 Therefore, it cost 45 percent more to provide equally good housing when the housing authority found the units and negotiated the rent than when tenants found their own units and negotiated the rent.

What accounts for the large differences in the total cost of providing equally good housing under programs of tenant-based and unit-based assistance? Among the plausible explanations are the absence of a financial incentive for good decisions on the part of civil servants who operate public housing,

^{8.} The administrative cost of the housing allowance program was about 15 percent of the total cost. This implies that landlords of housing allowance recipients received market rents for their units. Subsequent research has indicated that this is a common characteristic of all tenant-based rental assistance (Leger and Kennedy 1990; Wallace et al. 1981; Weinberg 1982).

the excessive profits that inevitably result from allocating subsidies to selected developers of private subsidized projects, the resources that developers devote to securing the subsidies, and the distortions in usage of inputs resulting from the subsidy formulas. A special case of the latter is that unit-based assistance is usually tied to the construction of new units. The least-expensive approach to improving the housing conditions of low-income households involves heavy reliance on upgrading the existing housing stock, the primary mechanism through which tenant-based assistance achieves this goal.

The results concerning the cost-effectiveness of different housing programs illustrate the virtue of substantial reliance on market mechanisms for achieving social goals, especially the virtue of forcing sellers to compete for the business of buyers. Under a program of tenant-based assistance, only suppliers who provide housing at the lowest cost given its features can remain in the program. If the property owner attempts to charge a voucher recipient a rent in excess of the market rent, the tenant will not remain in the unit indefinitely because she can move to a better unit without paying more for it. Under programs of project-based assistance, suppliers who receive payments in excess of market rents for their housing can remain in the program indefinitely because their tenants would lose their subsidies if they moved. These suppliers have a captive audience.

Number of Households Served

The difference in cost-effectiveness between tenant-based and unit-based housing assistance has major implications for the number of households that can be served with the current budget. If we compare programs of tenant-based and unit-based assistance that provide recipients with equally good housing for the same rent, the tenant-based programs will serve many more families with a given budget. Indeed, as will be shown later, it is possible to design a tenant-based voucher program that would serve more families in each major group and more of the poorest families of each size without spending more

money. Therefore, many eligible families and the taxpayers who want to help them will gain if tenant-based assistance replaces unit-based assistance.

The magnitude of the gain from shifting from unitbased to tenant-based rental assistance would be substantial. Even the smallest estimates of the excess costs of unit-based assistance imply that shifting ten families from unit-based to tenant-based assistance would enable us to serve two additional families. Since the federal government provides unit-based rental housing assistance to more than 5 million families, a total shift from unit-based to tenant-based assistance would enable us to serve at least 1 million additional families with no additional budget. The most reliable estimates in the literature imply much larger increases in the number of families served. For example, the results in Wallace et al. (1981) imply that tenant-based vouchers could have provided all of the families who participated in the Section 8 New Construction program with equally good housing for the same rent, and that those vouchers could have served at least 72 percent more families with similar characteristics equally well without any additional budget. In an analysis of the effect on program participation of replacing HUD's low-income housing programs with an entitlement housing voucher program that has the same cost to taxpayers, Olsen and Tebbs (in progress) find that the voucher program would serve an additional 2.4 million families. Section 5 of this paper summarizes the methods and presents some results of that research.

Household Choice

In addition to providing equally desirable housing at a lower cost, tenant-based assistance allows each recipient to occupy a dwelling unit with a combination of characteristics preferred to the specific dwelling unit offered under a program of unit-based assistance, without adversely affecting tax-payer interests. All low-income housing programs have minimum housing standards that reflect the interests of taxpayers in ensuring that low-income families live in adequate housing. Unit-based assis-

tance forces each family to live in a particular unit in order to receive a subsidy, greatly restricting recipient choice among units meeting minimum housing standards, whereas tenant-based assistance allows recipients to occupy any unit meeting the program's minimum housing standards. Units that meet the program's standards and that are affordable to assisted families differ greatly with respect to their condition, amenities, size, neighborhood, and location. Each family will choose the option that best suits its tastes and circumstances. When its circumstances change markedly, the family can move to another unit. Since all of these units are adequate as judged by the program's minimum housing standards, restricting the family's choice further serves no public purpose. If the subsidy is the same, it is reasonable to expect recipients of tenant-based assistance to be significantly better off than they would be with unit-based assistance.

4. Proposals to Shift Budget from Unit-Based to Tenant-Based Assistance

he available evidence on program performance has clear implications for housing policy reform. To serve the interests of tax-payers who want to help low-income families with their housing and the poorest families who have not been offered housing assistance, Congress should shift the budget for low-income housing assistance from unit-based to tenant-based housing assistance as soon as current contractual commitments permit, and should not authorize any new programs involving unit-based assistance.

The Clinton administration made detailed proposals to Congress to achieve this transition (HUD 1995), and Senator Dole supported the general concept during his presidential campaign against President Clinton (as reported by Guy Gugliotta, "Dole Urges Abolition of Public Housing," *Washington Post*, April 30, 1996, p. A05). The evidence on program performance that has accumulated in the dozen years since then has confirmed the wisdom of the main thrust of these proposals. It is time to refine and act on them. This section proposes some concrete steps to achieve the desired results.

Transitioning from Public Housing Assistance to Tenant-Based Assistance

First, the money currently spent on operation and modernization subsidies for public housing projects should be shifted gradually to provide tenant-based vouchers to public housing tenants. HUD provides housing authorities with about \$7 billion each year in operation and modernization subsidies. This is about one-fourth of the total HUD budget for low-income housing assistance. The evidence indicates that we can get more for this money by giving it to public housing tenants in the form of housing vouchers.

The Quality Housing and Work Responsibility Act of 1998 (QHWRA) made a small step in that direction by mandating the conversion of public housing projects to tenant-based assistance under certain circumstances, and by allowing it under other circumstances. However, it did not go nearly far enough to realize large gains. The following proposal will achieve these large gains in an orderly fashion.

The proposal would allocate to each housing agency the same amount of federal money as it would have received in operation and modernization subsidies under the current system so that no housing agency can argue against the proposal on the grounds that it would have less with which to serve its clients. With one caveat mentioned below, it would require every local housing agency to offer each current public housing tenant the option of a portable housing voucher or remaining in its current unit on the previous terms. The latter provision ensures that no public housing tenant is harmed by the legislation. Families that accept a voucher would benefit from it. They would be able to move to housing that they prefer to their public housing units. These vouchers would be funded from each agency's current public housing budget. Housing agencies would be allowed to charge whatever rent the market will bear for the units vacated by families that accept the voucher offer, and sell any of their projects to the highest bidder. This would generate the maximum amount of money to operate and modernize their remaining projects.

Given the difficulty of predicting the consequences of these far-reaching changes, it would be desirable to start with a demonstration program involving public housing authorities willing to implement these proposals for a randomly selected subset of their public housing projects. This demonstra-

^{9.} See Weicher (1997) for a detailed analysis of proposals for vouchering out unit-based assistance.

tion would produce evidence on the effects of the proposals, and would provide useful information for modifying them to avoid unforeseen negative consequences and to achieve better outcomes. The housing authorities that volunteer should receive additional administrative funds to compensate for the extra workload associated with this radical transformation of public housing, and HUD's Office of Policy Development and Research should receive funding for the research component of the demonstration so that we learn from the experiences of these pioneering housing authorities.

The remainder of this subsection spells out the details of the proposal and some options of the basic plan. It also anticipates some problems and discusses possible solutions to them.

The most important requirement of the proposal is that each housing agency must offer a housing voucher to each family currently living in a public housing project. The payment standards for families of each size (that is, the subsidy to a family with zero adjusted income) need not be the payment standards of the regular Section 8 Voucher program.¹⁰ To ensure that housing authorities can pay for these proposed vouchers with the money available, payment standards for families of different sizes should be set to use the housing agency's entire public housing budget in the highly unlikely event that all public housing tenants accepted vouchers. A set of payment standards that satisfies this criterion is easily calculated. Evidence from HUD's latest Picture of Subsidized Households (HUD 2000) indicates that the national average payment standard in the regular voucher program was \$653 a month in 2000, and it would have been \$628 a month in the proposed voucher program for public housing tenants. However, this small average difference does not preclude large differences for individual housing authorities.

It is important to realize that this proposal would

not lead to an immediate mass exodus from public housing. The results of the HUD-funded Moving to Opportunity (MTO) for Fair Housing Demonstration program show that public housing projects would retain the overwhelming majority of their tenants, at least initially.11 The families eligible to participate in the experiment lived in public housing projects in census tracts where the poverty rate exceeded 40 percent prior to the experiment. Only one-fourth of the families in the projects involved signed up to participate in the experiment within the specified time limit (Goering et al. 1999, p. 32). About 62 percent of the families offered regular Section 8 vouchers as an alternative to staying in their public housing unit used the voucher and left public housing (Orr et al. 2003, p. 26). This surely exceeds the fraction of all public housing tenants that would accept a regular Section 8 voucher if for no other reason than public housing tenants in lower poverty neighborhoods live in better neighborhoods. Nationally, about one-third of public housing tenants live in neighborhoods with poverty rates as high as those experienced by MTO participants (Newman and Schnare 1997, Table 3). Public housing projects in better neighborhoods probably also are newer and provide better housing.

The proposal would not require housing agencies to sell their projects beyond what will be required under the regulations implementing the relevant QHWRA provisions. However, it would allow them to sell any of their projects to the highest bidder, and many housing agencies would surely choose to sell their worst projects. With uniform vouchers offered across all of a housing agency's projects, it is reasonable to expect that the fraction of all public housing tenants that accept the vouchers would be greatest in these projects. These are also the projects that would be the most expensive to renovate up to a specified quality level. They are the types of projects that have been demolished under the HOPE VI program and that Congress intended to voucher out under QHWRA. Thus, the proposal

^{10.} This proposal does not affect the regular Section 8 Housing Choice Voucher Program.

^{11.} See Orr et al. (2003) for a description of the experiment and a summary of its results to date.

is consistent with clear congressional intent in this regard.

To be sure, some public housing tenants could be hurt by the proposal. Some tenants might want to remain in the projects that housing authorities decide to sell, even if they sell their worst projects. Some of these tenants might later discover that they prefer their new housing to their current units, but that does not preclude initial opposition to the sale. In evaluating this proposal, the losses to certain tenants must be weighed against the gains to other tenants. It is difficult to justify renovating structures that reach a certain level of obsolescence and dilapidation, and the Congress has made a policy decision to tear down the worst public housing projects even if some tenants would like to remain in them. Through 2005, about seventy-eight thousand distressed public housing units had been torn down and another ten thousand had been slated for demolition under HOPE VI (HUD 2007). Others have been demolished with funding from other sources. The number of public housing units has declined by more than 223,000 since its peak in 1991.

To minimize the number of public housing tenants forced to move, Congress could place restrictions on housing authority discretion with respect to selling its projects. The decade-long struggle to write the regulations to implement the 1998 QHWRA provisions concerning the mandatory and voluntary vouchering out of public housing projects argues for simple and unambiguous legislative provisions. One possible restriction on housing authority discretion is to require at least a certain percent of the tenants in a building to approve the sale. The higher the percentage, the smaller the number of projects that could be sold. Another possibility is to limit the number of units sold to the number of units vacated by families that accept vouchers. Since it is not practical to sell individual units within buildings, this rule should not be applied separately to each building. If 350 families living in projects operated by a housing authority leave public housing with vouchers, the housing authority would be allowed to sell buildings with up to this many units.

When a project is sold, the remaining tenants in that project would be offered the choice between vacant units in other public housing projects and a housing voucher. The Uniform Relocation Act requires housing agencies to provide each family that is forced to move with relocation assistance. Although out-of-pocket moving expenses within a metropolitan area are modest (about \$250 a room in a locality with an average price level), the housing authority would have to spend whatever is required by the Uniform Relocation Act. Some of the proceeds of the sale would have to be used for this purpose. The rest of the sales proceeds could be used to improve the housing authority's other projects or to offer housing vouchers to additional families on its waiting lists.

Each year some current public housing tenants that have not accepted the proposed vouchers will move from their units without these vouchers. For example, some will get jobs that pay so much that they are no longer eligible for housing assistance, some single mothers will get married thus raising their household income and making them ineligible, and some will be offered a preferred unit in a private subsidized project or a regular Section 8 voucher. When this happens, public housing agencies should be required to offer the family at the top of its public housing waiting list the option of occupying the vacated unit on the standard terms or accepting one of the new housing vouchers. These requirements will expand the choices of families who are offered housing assistance and will ensure that there is no reduction in the number of subsidized families.

When a current public housing tenant accepts a voucher, or when a family on the public housing waiting list rejects a public housing unit in favor of a voucher, the proposal requires the housing authority to charge the highest rent that the market would bear for the unit. This will be greater than the rent paid by the previous tenant in almost all cases and hence provides additional revenue to housing agencies without additional government subsidies and without reducing the number of assisted families. It also will make the housing agency's revenue depend

in part on the desirability of the housing provided, thereby encouraging better maintenance of public housing units. This aspect of the proposed reform will almost surely lead to some unsubsidized families living in each public housing project. The fraction will be different in different projects and will increase over time.

Under current law, occupancy of vacated public housing units is limited to families eligible for low-income housing assistance. Given the socioeconomic characteristics of the families living in public housing and the condition, amenities, and locations of these projects, applying this occupancy restriction to new tenants paying market rents would surely have little impact. For a family of four, the upper-income limit for eligibility is 80 percent of the local median income of all families. It is unlikely that many families with higher incomes would want to live in most existing public housing projects.

Current law also requires that at least 40 percent of new tenants of public housing projects have incomes less than limits based on 30 percent of the area's median income. Meeting this requirement might reduce the maximum rent that the housing agency can charge for its vacated units, but this does not affect the proposal in any fundamental way. Each housing agency would simply charge the highest rent that the market will bear for its vacated units subject to satisfying the income-targeting requirement. This would lead to the same income targeting as the current system.

To promote economic integration in public housing projects, Congress might want to eliminate the income targeting rules for families that pay market rents for public housing units. Indeed, it might want to eliminate upper-income limits for these families. Under the proposal, the new occupants will receive no public subsidy, and so income targeting would serve no public purpose. Eliminating these requirements would promote economic integration in public housing projects without reducing the number of families that receive housing assistance.

Each year some public housing tenants that used

the proposed vouchers to leave their public housing units will give up these vouchers for the same reasons that some tenants leave public housing. A new voucher should be offered to a family on the public housing waiting list to replace each such family that leaves the program. This will ensure that the tax money spent on public housing will continue to support at least the same number of families.

Offering the voucher option to all tenants requires additional administrative resources. The revenues generated by renting some units at market rates might be more than adequate for this purpose. However, the reforms would yield such large benefits to so many low-income families that they easily justify additional administrative fees from the federal government.

The preceding proposals would benefit many current public housing tenants without greater cost to taxpayers. Only public housing tenants who initially want to remain in units in projects sold by housing authorities and who do not prefer their new situation after their displacement would be hurt. The public housing tenants that accept vouchers would obviously be better off because they could have stayed in their current units on the old terms. They would move to housing meeting HUD's housing standards that better suits their preferences and circumstances. Under the proposal, each housing agency would receive the same amount each year from the federal government as under the current system, and each would have the same assets, namely, the land and structures on which its projects are located. However, these assets would be better used, and the proposal would provide housing agencies with more money to better serve assisted families who remain in public housing. The additional money would come from selling projects and charging market rents for the units vacated by current public housing tenants. The proposal would greatly facilitate the sale of projects that are not worth renovating. The requirement that these projects must be sold to the highest bidder ensures that the land and structures would be put to their highest valued use and maximizes the money available to help lowincome families with their housing. It also avoids scandals associated with sweetheart deals. The dysfunctional public housing program of the twentieth century would wither, but public housing agencies would do a much better job in helping low-income families with their housing without spending additional money.

Transitioning from Private Subsidized Projects to Tenant-Based Assistance

The second broad proposal is that contracts with the owners of private subsidized projects should not be renewed. The initial agreements that led to the building or substantial rehabilitation of these projects called for their owners to provide housing meeting certain standards to households with particular characteristics at certain rents for a specified number of years. At the end of the use agreement, the government must decide on the terms of the new agreement and the private parties must decide whether to participate on these terms. A substantial number of projects have come to the end of their use agreement in recent years and many more will come to the end of theirs over the next decade. When use agreements are not renewed, current occupants are provided with other housing assistance, usually tenant-based vouchers. Up to this point, housing policy has leaned heavily in the direction of providing owners with a sufficient subsidy to induce them to continue to serve the low-income households in their projects. This has been the primary purpose of the Section 8 Loan Management Set-Aside, the Preservation Incentives, and the Mark-to-Market program (Hilton et al. 2004, pp. 1-2).

We should not repeat the mistakes of these programs by continuing to subsidize private housing projects. Instead, we should give their tenants portable vouchers and force the owners to compete for their business. It is important to realize that for-profit sponsors will not agree to extend the use agreement unless this provides at least as much profit as operating in the unsubsidized market provides. If the owner is offered a lower profit than in the unsubsidized market, the owner will leave

the program. Therefore, government subsidies can only err in the direction of providing excess profits to these private projects. We should leave the job of getting value for the money spent to the people who have the greatest incentive to do it: the recipients of housing assistance.

It is often argued that giving families that live in privately owned subsidized housing projects portable housing vouchers at the end of the use agreement will force them to move. This concern is certainly not valid unless the payment standard is below the market rent of the unit. If the payment standard is equal to the market rent, it would enable the family to continue to live in its current unit without devoting more of its income to rent, and it would offer the family options to its current unit that it might prefer. The evidence on the cost-effectiveness of renewing use agreements versus tenant-based housing vouchers indicates that offering such vouchers would reduce the taxpayer cost of assisting these families. The savings could be used to assist additional families.

Others argue that the failure to renew use agreements on privately owned subsidized projects reduces the number of affordable housing units. In fact, portable vouchers make many units affordable to low-income families since they can use these vouchers outside of designated housing projects. When use agreements are extended, the only unit that is made affordable to an assisted family living in the project is its current unit. Terminating use agreements does not change the number of households or the number of dwelling units and has no effect on the overall vacancy rate.

Phasing Out Current Production Programs and Avoiding New Production Programs

Third, the construction of additional public or private projects should not be subsidized. This involves terminating or phasing out current production programs and avoiding new production programs.¹²

The LIHTC is the largest active production program. It subsidizes more units than all of the other active production programs combined. The Tax Credit program is already the nation's second largest low-income housing program, and it is the fastest growing. In 2006, the tax credits themselves involved a tax expenditure of about \$4 billion. However, these projects received additional development subsidies from state and local governments (usually funded through federal intergovernmental grants) accounting for one-third of total development subsidies (Cummings and DiPasquale 1999, p. 299). Therefore, the total development subsidies were about \$6 billion a year. Furthermore, GAO (1997, p. 40) found that owners of tax credit projects received subsidies in the form of unit-based or tenant-based Section 8 assistance on behalf of 40 percent of their tenants. If the per unit cost of these subsidies was equal to the per unit cost of tenantbased housing vouchers in 2006, those subsidies would add more than \$4 billion a year to the cost of the tax credit program. Thus, the cost of the program to taxpayers was about \$10 billion in 2006.

The GAO results on the cost-effectiveness of the Tax Credit program, combined with the results of studies of similar earlier programs, suggest that money currently spent on the LIHTC would be better spent on expanding the Section 8 Voucher program. It might be argued that the GAO results are not sufficiently compelling to justify immediate termination of this program, and that the Tax Credit program is sufficiently different from older production programs to make evidence of their effects of little relevance for this decision. The GAO results are convincing enough, however, to justify rescinding the indexing of the tax credit for inflation and immediately launching a careful, independent analysis of the cost-effectiveness of the Tax Credit program. The reduced tax expenditures on the LI-

HTC could be diverted to a refundable tax credit for low-income first-time homebuyers, thereby offsetting to some extent the antihomeownership bias of the current system of low-income housing assistance (Olsen 2007).

Similar remarks apply to other active production programs. For example, no additional money should be allocated to HOPE VI. This program has been HUD's major production program over the past decade. It is an initiative within the public housing program under which some of the worst public housing projects have been torn down and replaced by new housing built at lower density on the same site. This program is an improvement over traditional public housing in that it avoids concentrating the poorest families at high densities in projects. However, the GAO study reveals that it is highly cost-ineffective compared with tenantbased vouchers that also avoid these concentrations. Therefore, the money that would have been spent on HOPE VI is better allocated to the much more cost-effective Section 8 Voucher program or added to the budget of each housing authority to operate its reformed public housing program.¹³ This shift in the budget for housing assistance would allow us to provide all of the families that would have lived in HOPE VI projects with rental units meeting minimum housing standards, and would allow us to assist tens of thousands of additional families that would otherwise live in deplorable housing.

It might be argued that this recommendation ignores the positive effect of HOPE VI projects on their neighborhoods. HOPE VI projects are much more attractive than the housing projects that they replaced, the density of the housing is much lower, families with higher incomes occupy some of the units built, and the most troublesome tenants are not allowed to return. Therefore, it is reasonable to

^{12.} Under the cover of dealing with the current housing crisis, the Housing and Economic Recovery Act of 2008 signed by the president on July 30, 2008, moved the current system in the opposite direction. It established an affordable housing trust fund financed by Federal National Mortgage Association (FNMA, or Fannie Mae) and Federal Home Mortgage Corporation (FHMC, or Freddie Mac) and increased subsidies delivered through low-income housing tax credits and tax-exempt housing bond authority.

^{13.} This money could be divided among public housing authorities using a formula that accounts for the size of their public housing program and the ages of its units.

expect HOPE VI projects to make neighborhoods in which they are located more attractive places to live. However, the same beneficial effect on the neighborhood could surely be achieved at a small fraction of the cost of HOPE VI redevelopment. For example, the old public housing project could be torn down, some of its land devoted to public facilities such as parks, and the rest sold to the highest bidder. Many alternative uses of the land would surely improve the neighborhood as much as HOPE VI redevelopment and have a much lower taxpayer cost. The savings could be used to provide housing vouchers to a larger number of low-income households than were served by the old public housing project, let alone the HOPE VI redevelopment of that project. Selling much of the land to the highest bidder would almost surely lead to private redevelopment that would improve the neighborhood, and this sale would generate additional revenue to provide vouchers to more households. Since public housing projects pay only a small fraction of full property taxes, it also would generate more revenue for local governments to deal with a host of problems.

Finally, there should be no new production programs. Congress should reject the administration's proposal for a tax credit to selected builders of housing for low-income homeowners modeled after the LIHTC.14 It also should reject the Millennial Housing Commission's (2002) proposals to create new programs of unit-based assistance such as tax incentives to preserve and expand the stock of existing units providing unit-based assistance, a new rental production program with a 100 percent capital subsidy, and elimination of limits on the amounts of Mortgage Revenue Bonds that states can issue to finance low-income housing projects. For the same reason, Congress should reject the National Housing Trust Fund Act of 2007 (H.R. 2895) until it is modified to direct the funds involved to tenant-based assistance. Launching a new construction program is particularly inappropriate when rental vacancy rates are at historic high levels.¹⁵ Any additional money for housing assistance should be used to expand the Housing Choice Voucher program.

^{14.} The administration's American Dream Program to provide a part of the down payment on a house for low-income families is not subject to the same criticisms. Since this program is well designed to benefit low-income families and increase their homeownership rate without creating other distortions, a good case can be made for it.

^{15.} See http://www.census.gov/hhes/www/housing/hvs/qtr307/q307tab1.html.

5. Justification and Design of an Entitlement Housing Voucher Program

he argument for replacing unit-based with tenant-based assistance is based on evidence on program performance. This section argues for an entitlement program of housing assistance based on taxpayer preferences. In combination, these arguments imply that we should have an entitlement program of tenant-based housing assistance.

Unlike other major means-tested transfer programs, housing assistance as mandated by the Housing Act of 1949 is not an entitlement, despite its stated goal of "a decent home and suitable living environment for every American family." Millions of the poorest families are not offered any housing assistance, while a smaller number of equally poor families receive large subsidies. For example, an assisted family with one child and an adjusted annual income of \$10,000 living in an area with the program's average payment standard would have received an annual housing subsidy of \$6,600 from the Housing Choice Voucher program in 2007 if it occupied an apartment renting for at least the payment standard. The majority of families with the same characteristics living in that locality would receive no subsidy from any low-income housing program. Furthermore, the majority of the poorest eligible families receive no assistance while many families with considerably greater income are assisted. This is not because the poorest families do not want assistance on the terms offered: the waiting lists of public housing authorities are long, would be much longer in many cases if they were open continuously for new applicants, and consist largely of families with extremely low incomes.16

It is difficult to reconcile these features of the low-income housing programs with plausible taxpayer preferences. How would taxpayers who want to help low-income families with their housing feel about dividing a fixed amount of assistance between two families that are identical in his or her eyes? Surely, almost all would divide the money equally between the two families.

Another strong argument for an entitlement housing assistance program for the poorest individuals and families is its effect on homelessness (Khadduri 2008). The homeless are the poorest of the poor. Almost all would be eligible for an entitlement housing voucher program that could be funded with the current budget for low-income housing assistance. Without extraordinary outreach, an entitlement program of housing assistance for the poorest individuals and families would eliminate homelessness except possibly for the chronic homeless—those who suffer from serious mental illness and substance abuse (Early and Olsen 2002).¹⁷ The results of the recently completed Welfare to Work Voucher Experiment provide further evidence of the power of housing vouchers to address homelessness (Abt Associates et al. 2006). Housing vouchers also have proven extremely effective in getting the chronic homeless off the streets, though it requires a proactive approach to reach these people. These individuals are much more willing to live in a regular apartment than in a homeless shelter, and it is much easier to deliver other services to them when they have a fixed address than when they live on the streets.¹⁸ One of the country's leading experts on homelessness argues that "the current system of providing temporary shelter in lieu of rental assis-

^{16.} This information is in the annual plans that public housing authorities submit to HUD each year. See http://www.hud.gov/offices/pih/pha/.

^{17.} Existing homeless shelters will certainly be a part of the solution to dealing with homelessness. Due to the time necessary to determine eligibility and find regular housing, an entitlement housing assistance program for the poorest households will not eliminate the desirability of some short-term facilities to house people who would otherwise live on the streets. Although we might want to fund them in a different manner, existing shelters would surely be among the short-term facilities used.

^{18.} This is the thrust of the promising Housing First approach. See http://www.endhomelessness.org/section/tools/housingfirst.

tance would appear to be relatively inefficient, since it is a less direct method of addressing the affordability gap and since, compared with independent housing, it carries such significant administrative and facility costs in addition to the social costs of disruptive shelter stays on families and children" (Culhane et al. 2007, p. 26).

The preceding argues strongly that a program of housing assistance should be an entitlement for the poorest families. To say that housing assistance should be an entitlement is not to say that it should be designed to ensure that all eligible families participate. It is inevitable that the participation rate will be less than 100 percent even in a well-designed entitlement housing assistance program. An entitlement housing assistance program should provide no subsidy to families with incomes at the upper limit for eligibility to avoid the inequity that results from offering families with incomes just below the upper income limit a higher standard of living than families with incomes just above it. This implies that families with incomes just above the income limit for eligibility will be eligible for small subsidies. In order to get this subsidy, they will have to occupy a unit meeting particular housing standards, spend time filling out paperwork and dealing with program administrators, and reveal personal information. These requirements are all inherent in operating a means-tested housing program. Furthermore, few enjoy accepting public or private charity. For all of these reasons, many eligible families will choose not to participate in an entitlement housing assistance program.

What would be the participation rate in an entitlement housing program? The participation rate in the tenant-based entitlement housing assistance programs operated in the 1970s in Green Bay and South Bend as a part of EHAP was about 32 percent. However, evidence from EHAP indicates clearly that participation depends on the generosity of the

subsidy, among other things. Studies of the success rate in the Section 8 Certificate and Voucher programs also accord with this commonsensical result (Finkel and Buron 2001). Thus, the participation rate would be different in different entitlement housing programs.

In ongoing revisions of their 2006 paper, Olsen and Tebbs have estimated the effects on the number of families served that would result from replacing HUD's major low-income housing programs with several entitlement housing voucher programs that have the same direct taxpayer cost. These estimates are based on data from HUD's administrative records and the 5 percent Public Use Microdata Sample from the 2000 Decennial Census.¹⁹ The participation rate in the entitlement program for families of each type is based on experience from the only entitlement housing assistance program that has been operated in the United States: the Housing Assistance Supply Experiment. Like the Supply Experiment, the entitlement programs analyzed by Olsen and Tebbs provide a subsidy conditional on occupying a unit meeting certain minimum housing standards. The magnitude of the subsidy is equal to a payment standard minus 30 percent of the recipient's adjusted income. Payment standards are larger for families whose size and composition justifies more bedrooms, and they are different in different localities. They are designed so that recipients who occupy units renting for the local payment standard live in equally good housing in all locations. The upper-income limit for eligibility is the lowest income at which the subsidy is zero. Unlike the current system, then, income limits are the same in real terms in all localities. In Washington, DC in 2000, the upper-income limits for the proposed program analyzed below would range from \$17,479 a year for families entitled to an efficiency apartment to \$49,642 a year for families entitled to eight bedrooms. For families entitled to two bedrooms, it is \$19,421.

^{19.} For reasons that are not important for present purposes, single nonelderly households are excluded from the calculations. It is assumed that these households continue to be served by current programs. The money used to serve them is not available for the entitlement program.

Tables 6 through 9 report the results for one of the entitlement voucher programs analyzed. There were 88.2 million households in the United States in 2000 who were not single nonelderly households. About 13.3 million (15 percent) of these households would have been eligible for the proposed entitlement program. It is estimated that 5.8 million households would have participated in it, so the participation rate would have been about 44 percent. This is a low participation rate compared with existing entitlement welfare programs. However, it is reasonable to expect that the participation rate in an entitlement housing voucher program would be lower than in programs that provide equally generous subsidies for other goods because many families must move in order to satisfy the program's minimum housing standards. Subsidies for other goods do not have this cost of participation. The participation rate would be even lower for an equally generous entitlement program of unit-based housing assistance because all families must move in order to participate.

Table 6 reports the percentage of all households of each size in each real income decile that are served by the HUD programs replaced by the proposed entitlement housing voucher program. Even in the

lowest real income decile, the highest fraction served is less than 40 percent.²⁰ Table 7 displays participation rates in the HUD programs replaced for various subgroups by real income and household size. The most striking feature of these results is that participation rates are much higher for blacks than for whites or Hispanics with the same real income and household size. Table 8 reports the number of households in each real income and household size category served by the HUD programs replaced and the estimated number that would be served by the entitlement housing voucher program. It indicates that more households of each size in the lowest two deciles would be served by the entitlement program. Table 9 summarizes the overall results. It indicates that the proposed entitlement program would assist about 2.4 million more families than the HUD programs replaced, and it would serve more families of each type—white, black, or Hispanic; elderly or nonelderly; and families living in metropolitan or nonmetropolitan areas. Other entitlement voucher programs with somewhat different parameters would produce the same general pattern but different numerical results.

TABLE 6
Participation Rates in HUD Programs Replaced in 2000 (percent)

Real income decile			House	ehold size			
	1	2	3	4	5	6+	All
1	32.6	28.3	38.5	35.5	30.1	26.2	32.2
2	8.0	8.9	15.6	16.0	15.8	16.3	11.0
3	0.7	1.4	3.3	3.8	4.3	5.0	2.2
4	0.1	0.2	0.5	0.7	1.0	1.3	0.4
5	0.0	0.0	0.1	0.2	0.2	0.4	0.1

Sources: HUD's Multifamily Tenant Characteristics and Tenant Rental Assistance Certification Systems.

Note: Since they account for only one-tenth of a percent of all participants, all tables exclude households with real incomes greater than the median. Table also excludes single nonelderly households.

^{20.} The percentage served by all low-income housing programs would be larger in each cell, but probably not that much larger for the lowest real incomes because the largest program not included in the calculations, the LIHTC, serves few extremely poor households who do not receive assistance from one of the HUD programs replaced.

TABLE 7
Subgroup Participation Rates in HUD Programs Replaced in 2000 (percent)

Household size

	1	2	3	4	5	6+	All
Group			First decile o	of real income			
White	30.3	21.9	31.7	29.7	26.7	25.8	27.8
Black	43.9	52.4	64.1	62.5	51.0	44.1	53.4
Hispanic	45.4	28.6	28.5	24.6	18.2	13.7	28.2
Elderly	32.6	17.0	14.9	14.9	12.2	8.5	30.0
Nonelderly	N/A	31.1	39.6	36.0	30.5	26.7	33.9
Metro	35.3	29.0	38.6	35.7	30.4	26.9	33.5
Nonmetro	24.3	25.7	38.1	34.8	28.9	22.5	27.8
			Second decile	e of real income	!		
White	7.7	6.6	11.7	12.5	13.9	15.6	8.7
Black	11.7	20.4	32.1	35.3	31.7	30.3	25.1
Hispanic	5.8	9.5	11.6	10.8	9.8	8.3	9.9
Elderly	8.0	6.1	6.2	8.0	8.1	6.6	7.4
Nonelderly	N/A	10.4	16.5	16.3	16.1	16.6	14.1
Metro	8.4	9.8	16.4	16.7	16.3	16.7	11.8
Nonmetro	6.8	6.4	12.8	13.6	13.7	14.0	8.7

Source: Author's calculations with Jeff Tebbs.

Note: Table excludes single nonelderly households. N/A = not applicable

TABLE 8

Number of Households Served by HUD Programs Replaced and Entitlement Housing Voucher Program with Same Taxpayer Cost, 2000

Real income decile Household size							
	1	2	3	4	5	6+	All
			Program	s replaced			
1	838,989	587,152	441,722	260,064	113,117	67,458	2,308,501
2	208,467	233,394	192,975	135,183	69,608	50,004	889,630
3	11,545	42,441	44,826	37,182	21,693	18,024	175,711
4	674	6,662	8,002	7,889	5,599	5,123	33,947
5	114	1,760	2,141	2,365	1,673	1,570	9,622
			Entitleme	nt program			
1	971,552	915,576	680,436	437,317	215,067	128,449	3,348,398
2	263,148	419,661	498,664	384,732	220,462	138,459	1,925,126
3	0	24,658	106,245	121,934	115,044	114,582	482,464
4	0	0	562	3,453	7,754	35,661	47,430
5	0	0	0	0	124	7,617	7,741
			Absolute	e increase			
1	132,563	328,424	238,714	177,253	101,951	60,992	1,039,897
2	54,682	186,267	305,689	249,549	150,854	88,455	1,035,496
3	-11,545	-17,783	61,419	84,753	93,352	96,559	306,753
4	-674	-6,662	-7,440	-4,435	2,155	30,538	13,483
5	-114	-1,760	-2,141	-2,365	-1,549	6,048	-1,881

Sources: Data from PUMS5 2000 Decennial Census; participation prediction equation from Housing Assistance Supply Experiment
Note: Single nonelderly households are excluded from the reform. Under the proposed entitlement voucher program, the payment standard for two-bedroom units is
adjusted across localities for differences in the rents of identical units. The payment standard for an efficiency is 90 percent of the two-bedroom payment standard. The
payment standard for a one-bedroom unit is 95 percent of the payment standard for a two-bedroom unit. For other numbers of bedrooms, standard adjustments in
HUD's Section 8 Voucher program are used.

TABLE 9
Summary of Effects of Proposed Reform on Number of Households Served, 2000

Group	Programs replaced	Entitlement voucher	Absolute increase	Increase (percent)
All	3,442,113	5,812,601	2,370,488	69
White	1,886,970	3,358,501	1,471,531	78
Black	1,408,461	1,541,332	132,871	9
Hispanic	452,415	1,196,278	743,863	164
Elderly	1,224,730	1,588,103	363,373	30
Nonelderly	2,197,383	4,224,498	2,027,115	92
Metro	2,767,870	4,737,794	1,969,924	71
Nonmetro	654,243	1,074,807	420,564	64

Source: Author's calculations with Jeff Tebbs.

The main argument against making housing assistance an entitlement is that it would be too expensive. Delivering housing assistance to all currently eligible families using the current mix of housing programs would almost surely greatly increase the amount spent on housing assistance, though this magnitude has not been estimated. However, we do not have to make more than 40 percent of the population eligible for low-income housing assistance. Furthermore, we can reduce the fraction of housing assistance delivered through programs that are cost-ineffective, and we can provide new recipients of housing assistance with smaller subsidies.²¹ If we

reduce the fraction of the population eligible for housing assistance, increase the fraction of families served by tenant-based assistance, or reduce the subsidy to new recipients under each housing program, the cost of an entitlement housing assistance program would be less than commonly assumed. Olsen and Tebbs's (in progress) analysis shows that it is possible to design an entitlement housing voucher program that serves many more households in each major group and more of the poorest households of each size without spending more money.

^{21.} It is extremely difficult to determine how many households are eligible for low-income housing assistance under the current system because housing programs have implicit as well as explicit income limits, different explicit limits for admission into a program and continued receipt of assistance, and implicit limits within a program that depend on the particular dwelling unit involved. The most common explicit limit for initial receipt of assistance is HUD's very low income limit based on 50 percent of the local median income. In 2005, about 29 percent of all households (owners and renters) had incomes less than these limits. The most common explicit limit for continued receipt of assistance is HUD's low income limit based on 80 percent of the local median income. About 47 percent of all households had incomes less than these limits. See HUD's Housing Affordability Data System at http://www.huduser.org/datasets/hads/hads.html.

6. Major Effects of Proposed Reforms

he proposals in this paper involve shifting money from one method of delivering housing assistance to another. Since taxpayers continue to pay the same taxes to support housing assistance, they gain only because people whom they care about are better served. This section summarizes the effects of these reallocations on low-income households.

One major effect of the proposed shifts is that millions of additional households will receive housing assistance. Evidence indicates that these households would live in better housing and neighborhoods, and they would have more money to spend on other goods (Abt Associates et al. 2006). In addition, millions of households that would have received assistance under the old budgetary allocation would live in housing and neighborhoods that they prefer to the units in housing projects that they would have occupied with the continuation of the current system. The MTO demonstration provides direct evidence on the effects of the public housing reform proposal for public housing tenants living in census tracts with high poverty rates (Orr et al. 2003). Since the vouchers involved in the public housing reform proposal in this paper are about as generous as regular Section 8 vouchers on average across all housing authorities, this evidence should provide a good indication of the proposed reform's average effects for this subset of public housing projects. On average, the families that accepted the regular Section 8 voucher in MTO moved into a better dwelling unit, a safer and nicer neighborhood, and a census tract with a much lower poverty rate.

The effect of the proposed reforms on the type of neighborhood occupied is not limited to public housing tenants living in high poverty neighborhoods. In general, families with tenant-based housing vouchers are much less likely to live in high poverty areas than families in subsidized projects, and they are much more dispersed geographically. Newman and Schnare (1997, Table 3) report that only 5.3 percent of voucher recipients live in census tracts where the poverty rate exceeds 40 percent. In contrast, 10.4 percent of occupants of private subsidized projects and 36.5 percent of public housing tenants live in such census tracts. Devine et al. (2003, p.12) find that more than 80 percent of census tracts in the fifty largest metropolitan areas have some recipients of Section 8 vouchers, whereas only 17 percent of census tracts have private subsidized projects and 8 percent have public housing projects. Based on the preceding information, it seems likely that the proposed reforms will promote economic integration in housing with its associated benefits.

7. Objections to Exclusive Reliance on Tenant-Based Assistance

enant-based rental assistance has a much lower total cost than any program of unit-based assistance for providing equally desirable housing, and it offers recipients much greater choice concerning the characteristics, neighborhood, and location of their housing. This makes a strong case for exclusive reliance on tenant-based assistance. Objections to this reform have been raised. This section addresses the main objections.

First, some are concerned that tenant-based assistance will not work well in markets with the lowest vacancy rates because these markets do not have enough affordable vacant apartments that meet minimum housing standards to house all families who are offered vouchers. In fact, it is not necessary for the number of vacant apartments that meet minimum housing standards and are affordable to voucher recipients to exceed the number of new and recycled vouchers available in order to use all vouchers available. Many families offered vouchers already occupy apartments meeting the program's standards. We do not need vacant apartments for these families. They can participate without moving. Other families who are offered vouchers live in housing that does not meet Section 8 standards. However, these apartments can be repaired to meet the standards. Similarly, vacant apartments that do not initially meet the program's standards can be upgraded to meet the standards. In short, a tenant-based voucher program increases the supply of apartments meeting minimum housing standards.

The evidence from the tenant-based Section 8 Certificate and Voucher programs illustrates these general points. One detailed analysis is based on data from a national random sample of thirty-three public housing authorities in 1993 (Kennedy and Finkel 1994). Thirty percent of all recipients outside New York City continued to live in the apart-

ments that they occupied prior to participating in the program (Kennedy and Finkel, p.15).²² Fortyone percent of these apartments already met the program's standards and 59 percent were repaired to meet the standards (Kennedy and Finkel, p. 83). About 70 percent of all recipients outside New York City moved to a new unit. About 48 percent of these apartments were repaired to meet the program's standards (Kennedy and Finkel, p. 84). The rest moved to vacant apartments that already met the standards. Therefore, the apartments occupied by about half of the families that received certificates and vouchers outside New York City during this period were repaired to meet the program's standards. The previously mentioned sources have similar results for New York City. In this city, only 31 percent of the apartments occupied by recipients had to be repaired to meet the program's standards.

The Housing Assistance Supply Experiment of the EHAP provides additional evidence on the ability of tenant-based vouchers to increase the supply of apartments meeting minimum housing standards even in tight housing markets. The Supply Experiment involved operating an entitlement housing allowance program for ten years in St. Joseph County, Indiana (which includes South Bend) and Brown County, Wisconsin (which includes Green Bay). About 20 percent of the families in the two counties were eligible to receive assistance (Lowry 1983, pp. 92-93). These sites were chosen because of the great difference in their vacancy rates in order to determine whether the outcomes of an entitlement housing allowance program depend importantly on this factor. At the outset of the Supply Experiment in 1973-1974, the vacancy rates in Brown County and St. Joseph County were 5.1 percent and 10.6 percent, respectively (Lowry, p. 53). In 2000, only 26 percent of the seventy-five largest metropolitan areas had vacancy rates less than the vacancy rate

^{22.} The authors analyzed New York City separately from the other housing authorities.

in Brown County, and only 20 percent had vacancy rates greater than the vacancy rate in St. Joseph County. Contrary to widely held expectations, the participation rate differed little between the two sites. Indeed, it was higher in the locality with the lower vacancy rate (Lowry, p. 122).

Data for analysis were collected during the first five years of the experiment in each site. During that period, about eleven thousand dwellings were repaired or improved to meet program standards entirely in response to tenant-based assistance and about five thousand families improved their housing by moving into apartments already meeting these standards (Lowry 1983, p. 24). The former represented more than a 9 percent increase in the supply of apartments meeting minimum housing standards. Tenant-based assistance alone produced a greater percentage increase in the supply of adequate housing in these localities in five years than all of the federal government's production programs for low-income families have produced in the past sixty-five years (Cutts and Olsen 2002, p. 232). The annual cost per household was less than \$4,000 in today's prices.

Some argue that the low success rates in the Section 8 Voucher program in areas with low vacancy rates implies that the available vouchers cannot be used in these areas and hence new construction must be subsidized in order to serve additional low-income households. However, it is important to distinguish between a housing authority's voucher success rate and its voucher utilization rate. An authority's success rate is defined as the percentage of the families authorized to search for a unit that receive a subsidy by occupying a unit meeting the program's standards within the housing authority's time limit. Its utilization rate is the fraction of all vouchers in use.

An authority's success rate depends on many factors, including the local vacancy rate. One careful study of success rates (Kennedy and Finkel 1994) indicates that, among localities that are the same with respect to other factors, those with the lowest vacancy rates have the lowest success rates. Obviously, it is more difficult to locate a suitable unit when the vacancy rate is low. However, housing market tightness does not explain most of the variation in success rates. Success rates also vary with family characteristics and program parameters. For example, families that are eligible for larger subsidies due to lower incomes or higher payment standards have a higher success rate, presumably because they have a greater incentive to find a unit meeting the program's standards.

For many years, public housing authorities have overissued vouchers and thereby achieved high utilization rates despite low success rates. By overissuing vouchers early in the year and adjusting the recycling of the vouchers that are returned by families who leave the program late in the year, housing authorities are able to come close to using their voucher budget. Their ability to use the money allocated to them is further enhanced by federal regulations that allow housing authorities to exceed their voucher budgets in a given year by modest amounts using their reserves and borrowing against next year's allotment. According to HUD's Fiscal 2004 Performance and Accountability Report (HUD 2004), the voucher utilization rate was 98.5 percent in that year.²³

Although it is true that some families who are offered vouchers do not find housing that suits them and meets the program's standards within their housing authority's time limit, other eligible families in the same locality use these vouchers. This indicates clearly that the problem is not a lack of vacant apartments that meet program standards and

^{23.} Although housing authorities could achieve a voucher utilization rate close to 100 percent each year by adjusting the extent to which they overissue vouchers, they have not always done so. Like others, directors of housing authorities respond to incentives and disincentives. In recent years, they have faced disincentives that have led to lower voucher utilization rates. Sard and Coven (2006) analyze the effect of proposed changes in federal regulations intended to induce housing authorities to use all of their vouchers.

are affordable to voucher recipients or apartments whose owners are willing to upgrade them to meet program standards. In the tightest housing markets, these apartments are more difficult to locate for everyone; families who do not receive subsidies also have trouble locating apartments in tight housing markets. The real issue is not whether tenant-based vouchers can be used in all market conditions, but rather whether it would be better to use new construction or substantial rehabilitation programs in tight housing markets. Evidence from the 2001 and 2002 reports from the GAO (2002) study indicates that tenant-based vouchers are more cost-effective than production programs even in markets with low vacancy rates. Another key question is which type of assistance gets eligible families into satisfactory housing faster. If the choice is between authorizing additional vouchers or additional units under any construction program, the answer is clear: tenantbased vouchers get families into satisfactory housing much faster than any construction program even in the tightest housing markets. By overissuing vouchers, housing agencies can put all of their vouchers to use in less than a year in any market conditions. No production program can hope to match this speed.

The second major objection to the exclusive reliance on tenant-based assistance is that new construction promotes neighborhood revitalization to a much greater extent than does tenant-based assistance. The evidence suggests that there is little difference between housing programs in this regard.

The evidence from the EHAP is that even an entitlement housing voucher program will have modest effects on neighborhoods. The small literature on the Section 8 Voucher program confirms these findings for a similar nonentitlement program (Galster, Tatian, and Smith 1999; Lowry 1983, pp. 205–17). These programs result in the upgrading of many existing dwellings, but this is almost surely concentrated on their interiors.

It is plausible that a new subsidized project built at low density in a neighborhood with the worst housing and poorest families would make that neighborhood a more attractive place to live for some years after its construction. To compare the performance of different programs, however, it is important to consider the magnitude of neighborhood upgrading across all projects under each program over the life of these projects, the identity of the beneficiaries of this upgrading, and the extent to which upgrading of one neighborhood leads to the deterioration of other neighborhoods.

The primary beneficiaries of neighborhood upgrading will be the owners of nearby properties. Since the majority of the poorest families are renters, it is plausible to believe that most of the housing units surrounding housing projects located in the poorest neighborhoods are rental. Therefore, if a newly built subsidized project makes the neighborhood a more attractive place to live, the owners of this rental housing will charge higher rents and the value of their property will be greater. Since the occupants of this rental housing could have lived in a nicer neighborhood prior to the project by paying a higher rent, they are hurt by its construction. The poor in the project's neighborhood will benefit from the neighborhood upgrading only to the extent that they own the property surrounding the project.

With the passage of time, the initial residents will leave the improved neighborhood and others who value a better neighborhood more highly will replace them. Therefore, housing programs involving new construction will shift the location of the worst neighborhoods to some extent. The aforementioned possibilities are rarely recognized in discussions of housing policy, let alone studied.

What has been studied is the extent to which projects under various housing programs affect the desirability of the neighborhood. If a housing project makes its neighborhood a better place to live, it will increase neighborhood property values. Most existing studies find small positive effects on neighborhood property values on average for some programs and small negative effects for others (Ericksen and Rosenthal 2007; Galster, Smith, et al. 1999,

Chapter 4; Lee, Culhane, and Wachter 1999). The excellent study by Schwartz and colleagues (2006) is an exception. They find that a number of construction and rehabilitation programs in New York City have substantial positive effects on neighborhood property values. However, the weight of the evidence still favors the view that no federal housing program has substantial effects on neighborhood property values on average across all of its units.

Third, it is often argued that we need to subsidize the construction of housing projects because some types of individuals and families—especially the elderly, persons with disabilities, and large families—have difficulties using housing vouchers. In fact, the tenant-based voucher program serves many families of these types. According to HUD's "Resident Characteristics Report," 45 percent of families in this program include either an elderly member or a member with a disability, and about 25 percent of the families served have at least four members. Thus, the objection that housing vouchers cannot be used to serve the disabled, elderly, and large families is counterfactual.

It is true that public housing and private subsidized projects tend to serve the elderly to a greater extent than the voucher program. About 31 percent of the families in public housing include an elderly person, as opposed to 18 percent in the voucher program. Public housing serves more elderly families (and hence smaller families) primarily because neighborhood opposition to the building of projects for the elderly has been much less than opposition to building projects for families with children. As a result, a disproportionate number of public housing projects were built for the elderly. Similarly, private projects serve more elderly people, perhaps because they tend to be better tenants—they generally do less damage to the property, they generally do not disturb their neighbors, and so on. The owners of these projects are not paid more to serve tenants who are more troublesome.

However, housing vouchers serve persons with disabilities and large families to a greater extent.

Around 33 percent of households in public housing include a disabled person, compared with 38 percent in the voucher program. Only 18 percent include at least four members, as opposed to 25 percent in the voucher program. The qualitative difference between vouchers and privately owned subsidized projects is similar.

Furthermore, if Congress felt that the current Section 8 Voucher program underserved certain groups, it could easily modify the program to increase the number of families of these types served using targeting requirements similar to the income targeting requirements that apply to HUD's programs. The income targeting regulations require that at least a certain fraction of new recipients have extremely low incomes according to HUD's definition of this term. Participation in the voucher program by members of a group also could be increased by increasing the maximum subsidy available to these households and thereby increasing their success rate (Finkel and Buron 2001).

Finally, it has been argued that economies of scale in providing services other than housing to the elderly and disabled justify project-based housing assistance. Although the magnitude of these scale economies has not been studied, their existence is beyond doubt. They explain the existence of housing combined with assisted care in the unsubsidized market.

This does not argue, however, for project-based housing assistance: these economies can be achieved under a system of tenant-based housing assistance. The scale economies involved are surely smaller than scale economies in the provision of services in nursing homes. Medicaid subsidizes low-income individuals to live in nursing homes. This is not done by subsidizing their construction and requiring recipients to live in particular nursing homes in order to receive assistance. Instead, eligible individuals can choose any nursing home that meets the program's minimum standards for care and charges less than the program's maximum payment. Private business firms and not-for-profit organizations

have built nursing homes in response to the demand created by this subsidy. A similar approach could be used to provide low-income housing assistance to individuals with special needs. These individuals could be offered vouchers that are more generous on the condition that they live in a building that provides the desired extra services. As is the case under current housing programs that provide extra services, the extra subsidy would not have to be sufficient to pay for them. Under HUD's Section 202 Supportive Housing for the Elderly program, project owners are required to provide certain supportive services. However, the program places an upper limit of \$15 per unit per month on the amount of the HUD subsidy and standard tenant rent that can be used for supportive services. The bulk of the money for supportive services comes from other public and private sources. HUD's Section 811 Supportive Housing for Persons with Disabilities is a similar program. As in the case of these programs, the purpose of the more-generous voucher subsidy is to pay for the staff needed to arrange and maintain the subsidies for supportive services.

This section has addressed the major objections to exclusive reliance on tenant-based housing assistance. It argues that a tenant-based voucher program can be used to get recipients into adequate housing faster than production programs even in the tightest housing markets, in part because it rapidly increases the supply of adequate housing. We do not need production programs for this purpose. Production programs have not had a perceptibly greater effect on neighborhood revitalization than tenant-based vouchers. The housing voucher program serves well the disabled, elderly, and large families. If we want to serve more families of these types in a nonentitlement housing voucher program we can do so easily with targeting rules. In an entitlement program, that goal is easily achieved with higher subsidies. Finally, achieving economies of scale in the provision of services other than housing to recipients of low-income housing assistance can be achieved as easily with tenant-based as with project-based housing assistance.

8. Conclusion

iven the many competing demands on the federal budget in coming years, the question is, how can we continue to serve current recipients equally well and serve some of the poorest families who have not yet been offered assistance? The answer is that we must use the money available more wisely.

The stated goal of the Housing Act of 1949 is "a decent home and suitable living environment for every American family." We can move closer to this goal by transferring funds from less cost-effective methods for delivering housing assistance to the most cost-effective approach and providing smaller subsidies to new recipients of housing assistance than received by current recipients. Research on the effects of housing programs shows that we can serve current recipients equally well (that is, provide them with equally good housing for the same rent) and serve many additional families by shifting resources from unit-based to tenant-based assistance. We should learn from our past mistakes and not heed the call for new production programs. Indeed, we should go farther and terminate current production programs, and disengage from unit-based assistance to existing apartments as soon as current contractual commitments permit.

In assessing the political feasibility of the type of fundamental reform considered in this paper, it is important to realize that this reform need not be implemented overnight. A politically feasible reform would involve a transition that does not harm, or even benefits, the overwhelming majority of current recipients of low-income housing assistance. For example, public housing tenants could be offered a choice between housing vouchers and staying in their current units on the same terms. This will benefit some without hurting others. Current recipients of Section 8 vouchers could be allowed to receive the generous subsidies that are now offered by the program while new recipients receive less generous subsidies so that more households can be served. Reform also must honor legal commitments. For example, payments on current terms will be provided to owners of private subsidized projects until the end of their use agreements. Occupants of these projects will not be offered vouchers until that time, and they might be provided with relocation assistance if they decide to move.

References

- Abt Associates Inc., Gregory Mills, Daniel Gubits, Larry Orr,
 David Long, Judie Feins, Bulbul Kaul, Michelle Wood, Amy
 Jones & Associates, Cloudburst Consulting, et al. 2006.

 Effects of Housing Vouchers on Welfare Families. Washington,
 DC: U.S. Department of Housing and Urban Development.

 hsgvouchers.html
- Carliner, Michael S. 1998. "Development of Federal Homeownership 'Policy." *Housing Policy Debate* 9(2): 299–321.
- Culhane, Dennis P., Stephen Metraux, Jung Min Park, Maryanne Schretzman, and Jesse Valente. 2007. "Testing a Typology of Family Homelessness Based on Patterns of Public Shelter Utilization in Four U.S. Jurisdictions: Implications for Policy and Program Planning." Housing Policy Debate 18: 1–28.
- Cummings, Jean L., and Denise DiPasquale. 1999. "The Low-Income Housing Tax Credit: An Analysis of the First Ten Years." *Housing Policy Debate* 10: 251–307.
- Cutts, Amy Crews, and Edgar O. Olsen. 2002. "Are Section 8 Housing Subsidies Too High?" *Journal of Housing Economics* 11: 214–43.
- Devine, Deborah J., Robert W. Gray, Lester Rubin, and Lydia B. Taghavi. 2003. *Housing Choice Voucher Location Patterns*. Washington, DC: U.S. Department of Housing and Urban Development.
- Early, Dirk W., and Edgar O. Olsen. 2002. "Subsidized Housing, Emergency Shelters, and Homelessness: An Empirical Investigation Using Data from the 1990 Census." Advances in Economic Analysis & Policy 2: 1–34.
- Ericksen, Michael D., and Stuart S. Rosenthal. 2007. "Crowd Out, Stigma, and the Effect of Place-Based Subsidized Rental Housing." Unpublished (September). http://faculty.maxwell.syr.edu/rosenthal/
- Finkel, Meryl, and Larry Buron. 2001. Study on Section 8 Voucher Success Rates. Volume I. Quantitative Study of Success Rates in Metropolitan Areas. Washington, DC: Office of Policy Development and Research, U.S. Department of Housing and Urban Development.
- Finkel, Meryl, Donna DeMarco, Deborah Morse, Sandra Nolden, and Karen Rich. 1999. Status of HUD-Insured (or Held)

 Multifamily Rental Housing in 1995: Final Report. Cambridge,
 MA: Abt Associates Inc. (May).
- Galster, George, Robin E. Smith, Peter A. Tatian, and Anna M. Santiago (with Mary Cunningham and Charlene Y. Wilson). 1999. Assessing Property Value Impacts of Dispersed Housing Subsidy Programs: Final Report. Washington, DC: The Urban Institute (May).
- Galster, George C., Peter Tatian, and Robin Smith. 1999. "The Impact of Neighbors Who Use Section 8 Certificates on Property Value." Housing Policy Debate 10: 879–917.
- Goering, John, Joan Kraft, Judith Feins, Debra McInnis, Mary Joel Holin, and Huda Elhassan. 1999. *Moving to Opportunity for Fair Housing Demonstration Program: Current Status and Initial Findings*. Washington, DC: Office of Policy Development and Research, U.S. Department of Housing and Urban Development.
- Hilton, Richard, Charles Hanson, Joanne Anderson, Meryl Finkel, Ken Lam, Jill Khadduri, and Michelle Wood. 2004. Evaluation of the Mark-to-Market Program. Econometric, Inc.

- and Abt Associates Inc. Washington, DC: Office of Policy Development and Research, U.S. Department of Housing and Urban Development.
- Kennedy, Stephen D., and Meryl Finkel. 1994. Section 8 Rental Voucher and Rental Certificate Utilization Study. Washington, DC: Office of Policy Development and Research, U.S. Department of Housing and Urban Development.
- Khadduri, Jill. 2008. *Housing Vouchers Are Critical for Ending Family Homelessness*. Washington, DC: Homelessness Research Institute, National Alliance to End Homelessness (January).
- Lee, Chang-Moo, Dennis P. Culhane, and Susan M. Wachter. 1999. "The Differential Impacts of Federally Assisted Housing Programs on Nearby Property Values: A Philadelphia Case Study." *Housing Policy Debate* 10: 75–93.
- Leger, Mireille L., and Stephen D. Kennedy. 1990. Final Comprehensive Report of the Freestanding Housing Voucher Demonstration, Vol. 1 & 2. Cambridge, MA: Abt Associates Inc. (May).
- Lowry, Ira S. 1983. Experimenting With Housing Allowances: The Final Report of the Housing Assistance Supply Experiment.

 Cambridge, MA: Oelgeschlager, Gunn & Hain.
- Mayo, Stephen K., Shirley Mansfield, David Warner, and Richard Zwetchkenbaum. 1980. Housing Allowances and Other Rental Assistance Programs: A Comparison Based on the Housing Allowance Demand Experiment, Part 2: Costs and Efficiency. Cambridge, MA: Abt Associates Inc. (June).
- Mikesell, James J., Linda M. Ghelfi, Priscilla Salant, George Wallace, and Leslie A. Whitener. "Meeting the Housing Needs of Rural Residents: Results of the 1998 Survey of USDA's Single Family Direct Loan Program." Rural Development Research Report 91, United States Department of Agriculture, Economic Research Service, Washington, DC (December).
- Millennial Housing Commission. 2002. Meeting Our Nation's
 Housing Challenges: Report of the Bipartisan Millennial Housing
 Commission Appointed by the Congress of the United States.
 Washington, DC: U.S. Government Printing Office.
- Newman, Sandra J., and Ann B. Schnare. 1997. "And a Suitable Living Environment': The Failure of Housing Programs to Deliver on Neighborhood Quality." *Housing Policy Debate* 8: 703–41.
- Olsen, Edgar O. 2000. "The Cost-Effectiveness of Alternative Methods of Delivering Housing Subsidies." Working Paper 351, Thomas Jefferson Center for Political Economy, University of Virginia, Charlottesville, VA (December). htm#olsen
- ——. 2003. "Housing Programs for Low-Income Households." In *Means-Tested Transfer Programs in the U.S.*, ed. Robert Moffitt, National Bureau of Economic Research. Chicago: University of Chicago Press.
- . 2007. "Promoting Homeownership among Low-Income Households." Opportunity and Ownership Project Report No. 2, The Urban Institute, Washington, DC (August). http://www.urban.org/publications/411523.html
- ——. 2008. "Low-Income Housing Policy." In New Palgrave Dictionary of Economics, ed. Steven N. Durlauf and Lawrence E. Blume, 2nd ed. London: Macmillan. http://www.dictionaryofeconomics.com/resources/about_editors>

- Olsen, Edgar O., and David M. Barton. 1983. "The Benefits and Costs of Public Housing in New York City." *Journal of Public Economics* 20(April): 299–332.
- Olsen, Edgar O., and Jeffrey M. Tebbs. In Progress. "The Effect on Program Participation of Replacing Current Low-Income Housing Programs with an Entitlement Housing Voucher Program." Unpublished (January). http://www. virginia.edu/economics/olsen.htm>
- Orr, Larry, Joan Kraft, Jeffrey Kling, Judith Feins, Margery Austin Turner, Susan Popkin, Jeanne Brooks-Gunn, Greg Duncan, Lawrence Katz, and Jeffrey Liebman. 2003. Moving to Opportunity for Fair Housing Demonstration Program: Interim Impacts Evaluation. Washington, DC: HUD.
- Sard, Barbara, and Martha Coven. 2006. "Fixing the Housing Voucher Formula: A No-Cost Way to Strengthen the Section 8 Program." Center on Budget and Policy Priorities, Washington, DC (November 1).
- Schwartz, Amy Ellen, Ingrid Gould Ellen, Ioan Voicu, and Michael H. Schill. 2006. "The External Effects of Place-Based Subsidized Housing." Regional Science and Urban Economics 36(September): 679–707.
- Shroder, Mark, and Arthur Reiger. 2000. "Vouchers versus Production Revisited." *Journal of Housing Research* 11: 91–107.
- U.S. Department of Housing and Urban Development (HUD). 1974. *Housing in the Seventies*. Washington, DC: Government Printing Office.
- . 1995. HUD Reinvention: From Blueprint to Action.
 Washington, DC: Government Printing Office (March).
 . 2000. "Picture of Subsidized Households." Data set.

Report." <a href="mailto:reports/cforept.gov/offices/cfo/reports/cfo/repor

http://www.huduser.org/picture2000/index.html

. 2004. "Fiscal 2004 Performance and Accountability

- ——. 2007. 2006 Annual Report to Congress on HOPE VI.
 Washington, DC: U.S. Department of Housing and Urban Development.
- U.S. General Accounting Office (GAO). 1997. Tax Credits:

 Opportunities to Improve Oversight of the Low-Income Housing
 Program. GGD/RCED-97–55. Washington, DC: GAO.
- ——. 2001. Federal Housing Programs: What They Cost and What They Provide. GAO-01–901R. Washington, DC: GAO (July 18).
- ——. 2002. Federal Housing Assistance: Comparing the Characteristics and Costs of Housing Programs. GAO-02–76. Washington, DC: GAO.
- Wallace, James E., Susan Philipson Bloom, William L Holshouser, Shirley Mansfield, and Daniel H. Weinberg. 1981. Participation and Benefits in the Urban Section 8 Program: New Construction and Existing Housing, Vol. 1 & 2. Cambridge, MA: Abt Associates Inc. (January).
- Weicher, John. 1997. Privatizing Subsidized Housing. Washington, DC: American Enterprise Institute for Public Policy Research.
- Weinberg, Daniel H. 1982. "Housing Benefits from the Section 8 Program." *Evaluation Review* 6(February): 5–24.

Acknowledgments

The author appreciates financial support from the Searle Freedom Trust and many useful comments from Manasi Deshpande, Doug Elmendorf, Sandra Newman, Pascal Noel, Greg Russ, Barbara Sard, and Paige Shevlin.

Author

EDGAR O. OLSEN

Ed Olsen is a professor of economics at the University of Virginia where he has served as department chairman. His research on low-income housing policy has been published in professional journals such as the *American Economic Review, Journal of Political Economy, Journal of Public Economics, and Journal of Policy Analysis and Management*, and he wrote the chapter on low-income housing programs in the recent National Bureau of Economic Research volume on means-tested transfers in the United States.

Ed has been a consultant to HUD during six administrations, the GAO on their recent study comparing the cost-effectiveness of housing vouchers and construction programs, the New York State Office of Temporary and Disability Assistance on the shelter allowance in their TANF Program, and the U.S. Department of Justice in fair housing litigation concerning the public housing program. In recent years, he has testified on low-income housing policy before U.S. Senate Committee on Banking, Housing and Urban Affairs, the U.S. House Subcommittee on Housing and Community Opportunity of the Committee on Financial Services, and U.S. House Subcommittee on Federalism and the Census of the Committee on Government Reform.



ADVISORY COUNCIL

GEORGE A. AKERLOF

Koshland Professor of Economics, University of California, Berkeley and 2001 Nobel Laureate in Economics

ROGER C. ALTMAN

Chairman, Evercore Partners

HOWARD P. BERKOWITZ

Managing Director, BlackRock

Chief Executive Officer, BlackRock HPB Management

ALAN S. BLINDER

Gordon S. Rentschler Memorial Professor of Economics,

Princeton University

TIMOTHY C. COLLINS

Senior Managing Director and Chief Executive Officer,

Ripplewood Holdings, LLC

ROBERT E. CUMBY

Professor of Economics, School of Foreign Service,

Georgetown University

PETER A. DIAMOND

Institute Professor,

Massachusetts Institute of Technology

JOHN DOERR

Partner, Kleiner Perkins Caufield & Byers

CHRISTOPHER EDLEY, JR.

Dean and Professor, Boalt School of Law -

University of California, Berkeley

BLAIR W. EFFRON

Partner, Centerview Partners, LLC

HAROLD FORD, JR.

Vice Chairman, Merrill Lynch

MARK T. GALLOGLY

Managing Principal, Centerbridge Partners

MICHAEL D. GRANOFF

Chief Executive Officer, Pomona Capital

GLENN H. HUTCHINS

Founder and Managing Director, Silver Lake Partners

JAMES A. JOHNSON

Vice Chairman, Perseus, LLC and

Former Chair, Brookings Board of Trustees

NANCY KILLEFER

Senior Director, McKinsey & Co.

JACOB L. I FW

Managing Director and Chief Operating Officer,

Citigroup Global Wealth Management

ERIC MINDICH

Chief Executive Officer, Eton Park Capital Management

SUZANNE NORA JOHNSON

Senior Director and Former Vice Chairman

The Goldman Sachs Group, Inc.

RICHARD PERRY

Chief Executive Officer, Perry Capital

STEVEN RATTNER

Managing Principal, Quadrangle Group, LLC

ROBERT REISCHAUER President, Urban Institute

ALICE M. RIVLIN

Senior Fellow, The Brookings Institution and

Director of the Brookings Washington Research Program

CECILIA E. ROUSE

Professor of Economics and Public Affairs,

Princeton University

ROBERT E. RUBIN

Director and Senior Counselor, Citigroup Inc.

RALPH L. SCHLOSSTEIN

President, BlackRock, Inc.

GENE SPERLING

Senior Fellow for Economic Policy,

Center for American Progress

THOMAS F. STEYER

Senior Managing Partner,

Farallon Capital Management

LAWRENCE H. SUMMERS

Charles W. Eliot University Professor,

Harvard University

LAURA TYSON

Professor, Haas School of Business,

University of California, Berkeley

DANIEL B. ZWIRN

Managing Partner, D.B. Zwirn & Co.

DOUGLAS W. ELMENDORF

Director

THE BROOKINGS INSTITUTION
1775 Massachusetts Ave., NW, Washington, DC 20036
(202) 797-6279 • www.hamiltonproject.org

